

UvA-DARE (Digital Academic Repository)

How awareness changes the relative weights of evidence during human decision-making

de Lange, F.P.; van Gaal, S.; Lamme, V.A.F.; Dehaene, S.

DOI 10.1371/journal.pbio.1001203 Publication date 2011 Document Version Final published version

Published in PLoS Biology

Link to publication

Citation for published version (APA):

de Lange, F. P., van Gaal, S., Lamme, V. A. F., & Dehaene, S. (2011). How awareness changes the relative weights of evidence during human decision-making. *PLoS Biology*, *9*(11), e1001203. [e1001203]. https://doi.org/10.1371/journal.pbio.1001203

General rights

It is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), other than for strictly personal, individual use, unless the work is under an open content license (like Creative Commons).

Disclaimer/Complaints regulations

If you believe that digital publication of certain material infringes any of your rights or (privacy) interests, please let the Library know, stating your reasons. In case of a legitimate complaint, the Library will make the material inaccessible and/or remove it from the website. Please Ask the Library: https://uba.uva.nl/en/contact, or a letter to: Library of the University of Amsterdam, Secretariat, Singel 425, 1012 WP Amsterdam, The Netherlands. You will be contacted as soon as possible.

UvA-DARE is a service provided by the library of the University of Amsterdam (https://dare.uva.nl)

How Awareness Changes the Relative Weights of Evidence During Human Decision-Making

Floris P. de Lange¹*⁹, Simon van Gaal^{2,3,4}⁹, Victor A. F. Lamme^{2,5}, Stanislas Dehaene^{3,4,6,7}

1 Donders Institute for Brain, Cognition and Behavior, Radboud University Nijmegen, Nijmegen, the Netherlands, 2 Department of Psychology, University of Amsterdam, Amsterdam, the Netherlands, 3 Inserm, Cognitive Neuroimaging Unit, Gif-sur-Yvette, France, 4 Commissariat à l'Energie Atomique, Neurospin Center, Gif-sur-Yvette, France, 5 Cognitive Science Center Amsterdam, the Netherlands, 6 Université Paris-Sud 11, Orsay, France, 7 Collège de France, Paris, France

Abstract

Human decisions are based on accumulating evidence over time for different options. Here we ask a simple question: How is the accumulation of evidence affected by the level of awareness of the information? We examined the influence of awareness on decision-making using combined behavioral methods and magneto-encephalography (MEG). Participants were required to make decisions by accumulating evidence over a series of visually presented arrow stimuli whose visibility was modulated by masking. Behavioral results showed that participants could accumulate evidence under both high and low visibility. However, a top-down strategic modulation of the flow of incoming evidence was only present for stimuli with high visibility: once enough evidence had been accrued, participants strategically reduced the impact of new incoming stimuli. Also, decision-making speed and confidence were strongly modulated by the strength of the evidence for highvisible but not low-visible evidence, even though direct priming effects were identical for both types of stimuli. Neural recordings revealed that, while initial perceptual processing was independent of visibility, there was stronger top-down amplification for stimuli with high visibility than low visibility. Furthermore, neural markers of evidence accumulation over occipito-parietal cortex showed a strategic bias only for highly visible sensory information, speeding up processing and reducing neural computations related to the decision process. Our results indicate that the level of awareness of information changes decision-making: while accumulation of evidence already exists under low visibility conditions, high visibility allows evidence to be accumulated up to a higher level, leading to important strategical top-down changes in decision-making. Our results therefore suggest a potential role of awareness in deploying flexible strategies for biasing information acquisition in line with one's expectations and goals.

Citation: de Lange FP, van Gaal S, Lamme VAF, Dehaene S (2011) How Awareness Changes the Relative Weights of Evidence During Human Decision-Making. PLoS Biol 9(11): e1001203. doi:10.1371/journal.pbio.1001203

Academic Editor: Michael Posner, University of Oregon, United States of America

Received May 20, 2011; Accepted October 12, 2011; Published November 22, 2011

Copyright: © 2011 de Lange et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was supported by the Netherlands Organisation for Scientific Research (NWO-VENI awarded to FdL; NWO-Rubicon awarded to SvG) and by an ERC Advanced Investigator Grant (230355) to VAFL. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: floris.delange@donders.ru.nl

• These authors contributed equally to this work.

Introduction

Many decisions can be formalized as a process of accumulation of evidence over time, ultimately favoring one alternative over another [1,2]. Evidence accumulation models have successfully captured the neural dynamics of simple decisions in a visual motion categorization task [3] as well as more complex decisions in which discrete pieces of evidence need to be integrated [4]. Here we investigate whether accumulation of evidence is affected by the level of awareness of the information.

Visual subliminal priming studies have shown that perceptual [5–7], cognitive [8,9], motor [10], and executive [11,12] stages can all be influenced by subliminal information. Furthermore, the amount of priming and subliminal processing increases linearly with prime processing [13,14], suggesting that some stages of evidence accumulation can proceed without awareness. Also, relatively long-term effects of subliminal priming have sometimes been observed [15–18], suggesting that accumulation of unconscious information is possible. However, it is an open question whether and how awareness modulates the way evidence is accumulated during decision-making.

Contemporary models of subliminal information processing posit that subliminal information is marked by a lack of "global ignition" [19], meaning that it cannot enter into a global workspace system that allows it to be held in working memory and broadcasted to a variety of higher level neural processors. This lack of ignition may preclude "access" to the information. Therefore, awareness may be a necessary condition for biasing and modifying the sensory evidence in line with one's expectations and goals during decision-making.

In this study, we directly test the potential role of awareness in human decision-making using a previously described task in which participants have to accumulate sequentially presented pieces of evidence across an extended period of time. We previously observed a dependency of evidence accumulation on the amount of prior accumulated evidence: when prior evidence was already strong, participants weighted the newly incoming information much less than when prior evidence was weak and indecisive [20]. Here we hypothesize that this top-down modulation may depend on awareness. While accumulation may be possible irrespective of the level of awareness [15–18], it may appear qualitatively

Author Summary

When making a decision, we gather evidence for the different options and ultimately choose on the basis of the accumulated evidence. A fundamental question is whether and how conscious awareness of the evidence changes this decision-making process. Here, we examined the influence of sensory awareness on decision-making using behavioral studies and magneto-encephalographic recordings in human participants. In our task, participants had to indicate the prevailing direction of five arrows presented on a screen that each pointed either left or right, and in different trials these arrows were either easy to see (high visibility) or difficult to see (low visibility). Behavioral and neural recordings show that evidence accumulation changed from a linear to a non-linear integration strategy with increasing stimulus visibility. In particular, the impact of later evidence was reduced when more evidence had been accrued, but only for highly visible information. By contrast, barely perceptible arrows contributed equally to a decision because participants needed to continue to accumulate evidence in order to make an accurate decision. These results suggest that consciousness may play a role in decision-making by biasing the accumulation of new evidence.

different depending on awareness level. Specifically, if awareness is necessary for top-down biasing of information during decisionmaking, low-visible evidence may be accumulated in a linear fashion, i.e. adding and subtracting new information without regard to the history of prior accumulated evidence. Nonlinearities in evidence accumulation (e.g., giving less weight to new information under conditions of high certainty [20,21]), which are a more optimal decision strategy within a Bayesian decision-making framework [22], may be present only for highvisible evidence.

We tested this hypothesis by using a decision-making task in which a sequence of five arrows was presented at either high or low visibility (HV versus LV), by means of masking (Figure 1A). In a series of behavioral experiments, we established whether and how evidence is accumulated over time, depending on the level of awareness. We also assessed the relationship between accumulated evidence and subjective decision confidence for stimuli at both awareness levels. Finally, we tracked accumulation-related neural activity over time in the human brain for both types of information, using magneto-encephalography (MEG).

Results

Behavioral Markers of Evidence Accumulation Over Time

On each trial, participants (N=16) were presented with a stream of five arrows, each of which could point to the left or right with equal probability. Participants had to quickly decide on the overall direction of the arrows by pressing a button at the end of each stream with their left or right index finger, guessing if necessary (Figure 1A). Strength of the evidence could range from one (low evidence, e.g. two left and three right arrows) to five (high evidence, e.g. five right arrows, see evidence accumulation diagram in Figure 1B). Visibility of the arrows was manipulated by masking them with an effective "metacontrast" mask (leading to low visibility, LV) or with an equiluminant but less effective "pseudo" mask (leading to high visibility, HV; see Figure S1 for details) [14,23]. On each trial, all arrows were either LV or HV. Stimulus and mask duration were identical for LV and HV

conditions, allowing us to compare behavioral performance of evidence accumulation and the underlying neural responses without confounding stimulus visibility with basic task parameters [24]. A six-choice discrimination task performed after the main experiment confirmed that visibility of the arrows was much poorer on LV trials than on HV trials (31% correct versus 73% correct, Figure 1C; difference: p < 0.001, see Materials and Methods for further details).

In the decision-making task, behavioral performance was less accurate for LV than for HV trials (60% versus 81% correct, p < 0.001). Nevertheless, for both LV and HV trials, performance increased with increasing amount of evidence (both p < 0.001), and thus evidence was accumulated for both trial types. For HV trials, performance approached a ceiling level of 100% correct for the highest evidence levels. For LV trials, the increase in performance was linear, peaking at 73% correct on trials with five identical arrows (Figure 1D). There was a trend of a right-side bias for LV trials when all arrows pointed in the same direction (67% versus 79%: T = 1.78, p = 0.096). Although we had no a priori hypothesis for such a bias, the finding is in line with earlier psychophysical studies showing that choices can be strongly biased by hand-preference especially under conditions of uncertainty (as in the case of the LV stimuli) [25–27].

While evidence accumulation was present for both LV and HV stimuli, there were striking behavioral differences in terms of how evidence was accumulated between conditions. First, the speed of decision-making was modulated by the amount of accumulated evidence for HV ($p \le 0.001$) but not for LV information (p = 0.31), leading to a significant interaction (p = 0.025, Figure 1E). Second, the "impact" of each successive arrow on the final decision varied as a function of time and accumulated evidence only for HV trials. We defined the impact of an arrow as the extent to which the arrow changed the response proportion in the direction of the arrow (see Materials and Methods for details on the exact quantification procedure). We observed a monotonically increasing impact of arrows on the decision as a function of time for HV stimuli (p < 0.001), while this modulation of time was only marginal for LV stimuli (p = 0.07), leading to a significant interaction $(p \le 0.001$, Figure 1F). Moreover, for HV arrows, the influence of the last arrow, defined as the extent to which it changed the response probability in the direction of the arrow, decreased linearly with the amount of previously accumulated evidence: the larger the amount of accumulated evidence, the less influence the last arrow had on the decision (p < 0.001), as expected from a rational strategy of progressively disregarding the arrows once sufficient evidence is obtained (Figure 1B). This modulation of accumulation by prior evidence was absent for LV stimuli (p = 0.44), leading to a significant interaction (p < 0.001, Figure 1G).

Together, these results show that strategic effects on decisionmaking strongly depend on the awareness level of the stimuli. Interestingly, these results were not simply due to the possibility that, during HV trials, participants stopped performing the task after having observed a sufficient amount of arrows. A "logical counting" algorithm would not give any weight to the last arrow when two or four pieces of evidence had already been accumulated, since the last arrow cannot change the decision anymore. In our data, however, the last arrow did have a sizeable influence on the decision even when four pieces of evidence had already been accumulated (Figure 1G, red line, right data point).

We further explored the relationship between decision-making performance and subjective confidence in a new group of 16 participants; this time we additionally asked them to rate their confidence of having responded correctly after every trial on a 6point scale (1 = pure guess, 6 = 100% sure). Overall decision-



Figure 1. Task set-up and behavioral performance. (A) Task. Participants were shown a sequence of five arrows that were briefly shown and subsequently masked. Masks were constructed such that the arrows were either highly visible (HV) or near the threshold of awareness (low-visibility, LV). Participants had to decide whether the predominant direction of the arrow sequence was left or right. (B) Evidence accumulation diagram. Participants start with no evidence for either direction. Each incoming arrow moves the sum up or down in the diagram. Solid lines show the transitions for the HV example in panel A. States at which no decision can yet be made are highlighted in light-grey; states at which enough evidence is available for the decision are highlighted in dark-grey. (C) Individual performance during the six-choice discrimination task (Experiment 2) for HV and LV stimuli. (D) Decision-making performance as a function of accumulated evidence for HV and LV stimuli mumbers denote evidence for a left and right response, respectively (number of right-arrows minus number of left-arrows). (E) Reaction times as a function of each arrow on the final decision (i.e., the extent to which the direction of the arrow determined the decision) over time for HV and LV triauls. (G) Influence of the last arrow on the final decision as a function of the amount of previously accumulated evidence for HV and LV stimuli. (H) Confidence ratings of visibility for HV and LV arrows (Experiment 3). (J) Relationship between confidence level and performance for HV and LV stimuli (Experiment 3). (K) Priming strength of single HV or LV arrow as measured in a distinct masked priming task, in terms of error rate (left panel) and reaction times (right panel, Experiment 4).

doi:10.1371/journal.pbio.1001203.g001

making performance was similar as in the MEG environment (in terms of overall performance and increase in performance with increasing evidence). As expected, overall confidence for LV arrows was much lower than for HV arrows (1.9 versus 3.8: p < 0.001; Figure 1H). More interestingly, participants' confidence level for correct responses and incorrect responses was nearly equal for LV arrows (difference = 0.05, p = 0.093), but strongly dissociated for HV arrows (difference = 1.07, p < 0.001; Figure 1I), resulting in a significant interaction ($p \le 0.001$). Thus participants had little insight in their accuracy level when arrows were strongly masked (LV), but they could very well distinguish correct from error trials when arrows were only weakly masked (HV). Since the inability to perform second-order confidence judgments has been proposed as a marker of lack of awareness [28], the results confirm that awareness was strongly reduced on LV compared to HV trials. When directly correlating decision-making performance and confidence for the correct trials, there was a strong correlation between these measures for HV trials (r=0.77, p < 0.001), while there was only a weak and marginally significant correlation for LV arrows (r=0.23, p=0.087, Figure 1J). However, this was likely due to the restricted range of confidence during LV since most participants reported low confidence levels for LV trials. When the range was restricted to the lowest three confidence levels, there was in fact a significant correlation between decision-making performance and confidence also for LV trials (r=0.66, p < 0.001). Overall, the results revealed that participants had markedly reduced confidence in their decision making for LV arrows, but could still use the information to achieve above-chance performance on the decision task.

Finally, we tested whether there was a difference in "stimulus strength" between the LV and HV arrows. In all experiments described so far, the arrow stimuli themselves were identical between conditions, and the only difference was the efficacy of the mask. Therefore, theoretically, the bottom-up stimulus strength, i.e. the ability of the stimulus to automatically climb up the sensorimotor pathways, may be equal for both conditions, even though visibility was strongly dissociated. To test this notion directly, we assessed and directly compared the priming strength of the LV and HV stimuli. Sixteen participants performed a simple masked priming experiment in which they responded as fast as possible to the direction of the mask (whose external outline was changed to a left- or right-pointing arrow; see Figure S1 and [14]). The mask was preceded by a LV or HV prime arrow pointing into the same or the opposite direction as the target. Congruence of the arrow prime resulted in significantly shorter reaction times (RT) and lower error rates (ER) to the mask for both LV and HV primes (all p < 0.001). Crucially, this priming effect was not significantly different between LV and HV primes (RT priming effect: LV = 55 ms, HV = 50 ms, p = 0.13; ER priming effect: LV = 18%, HV = 17%, p = 0.39; Figure 1K), in line with earlier findings [29]. This shows that the bottom-up stimulus strength is equal for LV and HV arrows, and points to an interesting dissociation between the "direct" priming impact of a stimulus and its visibility and its long-term effect on decision-making.

Neural Markers of Visibility

Using MEG, we first investigated whether LV and HV arrows were processed differently in the human brain, irrespective of evidence or direction. We used a cluster-level statistic to establish the significance of differences between conditions. This method effectively controls the type I error rate in situations involving multiple comparisons (such as 275 sensors) by clustering neighboring sensor pairs that exhibit the same effect (see Materials and Methods for more details). A direct comparison of LV and HV arrows, collapsed across all five arrows, revealed that there was larger activity for HV than LV arrows over left frontal and fronto-central sensors (50–100 ms interval, $p_{cluster} = 0.018$ and $p_{\text{cluster}} = 0.016$, respectively). At a later interval, there was larger activity for HV than LV arrows over parietal (100-150 ms interval, $p_{\text{cluster}} = 0.001$) and occipital (150–300 ms interval, $p_{\text{cluster}} < 0.001$; Figure 2) sensors. A detailed time course analysis estimated the first point of significant difference at 55 ms for the frontal cluster, 125 ms for the parietal cluster, and 145 ms for the occipital cluster (Figure 2B). The early frontal difference between HV and LV arrows was present for all arrows except for the first arrow of the sequence (Figure S2A), while the occipital and parietal amplification for HV arrows was visible for each and every arrow (Figure S2B,C). Interestingly, there was a behavioral counterpart of the frontal asymmetry between the first and subsequent arrows: whereas the first arrow had equal effects on the decision for LV and HV arrows, there were large differences in the weight of the subsequent arrows on the decision (Figure 1F).

Neural Markers of Evidence Accumulation Over Time

Previously we identified an inverse relationship between parietal and central neural activity and the amount of accumulated evidence: when more evidence was accumulated, neural activity evoked by new incoming stimuli was attenuated [20] (see [30,31] for comparable results). This pattern is consistent with the strategy to reduce the weight of new evidence once substantial evidence has already been accumulated. Behavioral results indeed showed that the impact of the last arrow decreased with the total amount of previously accumulated evidence, but for HV arrows only (Figure 1G).

For the analysis of evidence accumulation in the MEG environment, we compared activity for LV and HV arrows that



Figure 2. Neural markers of visibility. (A) Topographical representation of overall activity engendered by HV (top row) and LV (middle row) arrows, as well as differences in activity between them (bottom row). Activity (differences) is shown at five intervals after the onset of the arrow, from 50–300 ms. Dots indicate significant clusters of larger activity for HV than LV stimuli (p<0.05, corrected for multiple comparisons). (B) Time course of fronto-central (left panel), parietal (middle panel), and occipital (right panel) clusters for HV and LV arrows. Time course of frontal activity was plotted for all sensors with significant differences in the 50–100 ms interval, the parietal activity time course for the significant parietal sensors in the 100–150 ms interval, and occipital activity for the significant occipital sensors in the 150–200 ms interval. The dotted lines and numbers indicate the latency (in ms) at which differences between conditions first became significant. doi:10.1371/journal.pbio.1001203.g002

had "low prior accumulated evidence" and "high prior accumulated evidence," averaged across the third to fifth arrow presentation (the first two arrow presentations are not taken into account since there is no differential amount of prior accumulated evidence until after the first two arrows have been presented). Low evidence consisted of trials with zero (for third and fifth arrow) or one (for fourth arrow) prior accumulated evidence at the onset of the arrow. High evidence consisted of trials with two (for third and fifth arrow), three (for fourth arrow), or four (for fifth arrow) prior accumulated evidence at arrow onset.

We found that when participants had high prior accumulated evidence, the newly incoming arrows evoked a smaller activity at right occipito-parietal and central sensors. Crucially, this phenomenon was significant only for HV arrows (central sensors: 150–200 ms interval, $p_{cluster} = 0.014$; occipito-parietal sensors: 250–300 ms interval, $p_{cluster} = 0.041$) (Figure 3A, top row), while there was only a non-significant trend for LV arrows in central sensors (150–200 ms interval, $p_{cluster} = 0.077$) (Figure 3A, middle row). This resulted in a significant difference between conditions over right occipito-parietal sensors (HV versus LV: 250–300 ms

4



Figure 3. Neural markers of accumulated evidence. (A) Topographical representation of larger activity for arrows when the amount of previously accumulated evidence was low (low evidence), compared to when the amount of previously accumulated evidence was high (high evidence). Other conventions as in Figure 2. (B) Time course of central and parietal clusters. Central activity is plotted for the cluster that showed significantly reduced activity when accumulated evidence is high for HV arrows in the 150–200 ms interval. Parietal activity is plotted for the cluster that showed significantly stronger effects of accumulated evidence for HV than LV arrows in the 250–300 ms interval.

doi:10.1371/journal.pbio.1001203.g003

interval, $p_{cluster} = 0.042$) (Figure 3A, bottom row). Whereas neural responses are collapsed across arrows in this figure, Figure S2B shows that this effect was robustly observed in response to individual arrow presentations preceded by low and high evidence (a difference defined only for the third to fifth arrow, since differences in amount of accumulated evidence only arise after two arrows have been presented).

Neural Activity Related to a Change of Evidence

Under conditions of purely linear addition and subtraction of information, the direction of the previous arrow should not influence how the current arrow is processed. However, previous studies have described an automatic influence of repetition compared to alternation during decision-making [32], and previously we also showed a large reduction in neural activity for repeated compared to non-repeated arrows under conditions of high visibility [20]. When directly contrasting "repeat" arrows (i.e. arrows that were preceded by an arrow with the same direction) with "change" arrows (i.e. arrows that were preceded by an arrow with the opposite direction), we observed a large neural activity reduction for arrow repetitions (neural responses are collapsed across arrows). For HV arrows, this reduction was visible at occipito-parietal (100–150 ms interval, $p_{cluster} = 0.030;$ 150– 200 ms interval, $p_{\text{cluster}} = 0.005$; 200–250 ms interval, $p_{\text{cluster}} =$ 0.022) and central (HV: 200–250 ms interval, $p_{\text{cluster}} = 0.019$) sensors (Figure 4A, top row). A similar effect was also observed for LV arrows at central sensors only (200-250 ms interval, $p_{\text{cluster}} = 0.022$) (Figure 4A, middle row), in line with earlier studies showing subliminal repetition suppression effects [9,33]. Nevertheless, a direct comparison between both conditions shows that this suppression effect was significantly larger for HV than LV arrows (HV versus LV: 100–150 ms interval, $p_{cluster} = 0.039$; 150– 200 ms interval, $p_{\text{cluster}} = 0.015$; 200–250 ms interval, $p_{\text{cluster}} =$ 0.05) (Figure 4A, bottom row). Examination of the neural response to each individual "change" and "repeat" arrows (defined only for the second to fifth arrow, since the first arrow does not have a preceding arrow) shows that this effect was robustly found whenever a new arrow was presented, with no tendency to decrease with time (Figure S2C). Further, restricting the LV analysis to the poorest perceivers who scored at chance level in the six-choice discrimination task (16.7%) showed that this effect was present equally robustly for these nine "poor perceivers" (see Figure S3).

a) Repeat>Change for HV, LV and the difference



Figure 4. Neural markers of change in evidence. (A) Topographical representation of reduced activity for arrow stimuli that were preceded by an arrow with the same direction ("repeat"), compared to arrow stimuli which were preceded by an arrow with the opposite direction ("change"). Other conventions as in Figure 2. (B) Time course of central and parietal clusters. Central activity is plotted for the cluster that showed significant differences for both LV and HV arrows in the 200–250 ms interval. Parietal activity is plotted for the bilateral cluster that shows a significantly larger difference for HV than LV arrows in the 150–250 ms interval.

doi:10.1371/journal.pbio.1001203.g004

Discussion

In a combination of behavioral and electrophysiological studies, we showed that while participants are able to accumulate evidence over time independently from the level of awareness of the evidence, there were marked differences between accumulation of low-visibility (LV) and high-visibility (HV) information, both in terms of brain activity and behavior. Although the amount of bottom-up information provided by a single HV or LV arrow was identical, as measured by priming (Figure 1K), the overall decision-making performance was much less accurate when based on LV evidence than on HV evidence (Figure 1D). More interestingly, decision-making speed was modulated by the amount of accumulated evidence, but only for HV stimuli (Figure 1E). Also, subjective confidence in decision making was markedly lower for LV than HV evidence (Figure 1H). Together, this suggests that while participants could accumulate LV evidence over time, there are qualitative differences in accumulation of evidence depending on the level of awareness of the sensory information.

We observed a strong top-down biasing effect of the amount of previously accumulated information, only for HV evidence: the impact of the last arrow stimulus on the final decision decreased linearly with the amount of previously accumulated evidence (Figure 1G). Interestingly, participants did not stop accumulating HV evidence altogether when they had accrued enough information for their decision: even when a large amount of evidence (4 units) had already been accrued for one of the two decisions, the last arrow still had an impact on the decision process, which was equally large as the impact of any of the LV stimuli. This suggests that participants did not adopt a fully "logical" or digital counting strategy (perhaps for lack of time, as arrows come in at a fast pace of one every 300 ms). Rather, on HV trials only, they attributed a weight to later arrows that was inversely related to the amount of already accumulated evidence.

These behavioral findings constrain the theoretical modeling of the task. The observed strategy is not predicted by simple linear accumulation models [34], since these would predict equal weighting of later arrows, independently of the amount of previously accumulated evidence. It is also not in line with a simple gain of accumulation from LV to HV stimuli, since this would result in overall larger weights of each arrow, but no differential modulation by time and prior accumulated evidence. Rather, the results suggest a more sophisticated mode of evidence accumulation, in which the update signal is scaled with respect to the previously accumulated evidence. This behavior arises naturally from Bayesian and sequential sampling (SPRT) models [20,21], where evidence is only accumulated up to a bound. Beyond this bound, further evidence no longer contributes to the decision, with two consequences: (1) on average, later evidence is given a smaller weight, especially when early evidence is strong and the bound is therefore likely to be reached; (2) response time accelerates in proportion to the likelihood of reaching the bound. Both of these properties accurately characterize the participants' behavior on HV trials.

Importantly however, this modulation of evidence accumulation by prior accumulated evidence was absent for LV stimuli, where the impact of each arrow was not dependent on temporal position or prior amount of accumulated evidence. Such a purely linear accumulation of evidence is exactly what is predicted from optimal Bayesian integration, assuming that the amount of available evidence is low and therefore the accumulated amount rarely reaches threshold (see [20], Figure 2C). This hypothesis can also explain why RT remained constant on LV trials, independently of total evidence: even when five arrows point in the same direction, the total accumulation would still remain below the decision threshold on most trials, thus always requiring a forced-choice decision.

Overall, the simplest theoretical model therefore is that LV and HV trials were processed through a similar accumulation-decision pathway, yet with LV trials yielding a much lower level of evidence and therefore remaining far from decision threshold. Conversely, full awareness of the stimuli may be necessary for their accumulated evidence to reach a decision threshold which enables strategic top-down biasing of later evidence accumulation based on the past accumulation.

Magneto-encephalographic (MEG) recordings lend further support to this view. They showed that, while initial perceptual processing was identical for LV and HV evidence, there was a late divergence between LV and HV, which could be seen ~ 145 ms after stimulus onset over occipital cortex (Figure 2). This late divergence between LV and HV stimuli has been observed earlier using different masking paradigms [35-37], and these findings are generally in good accordance with a feedback view of masking, in which initial processing in visual areas is intact but late amplification by feedback is disturbed [19,38-40]. Note, however, that our data do not allow us to make firm claims about the underlying mechanisms of metacontrast masking, as different explanations have also been put forward to explain the late sensory divergence (e.g., [37]). There was also a neural difference between LV and HV stimuli over left frontal and fronto-central sensors, which became significant as early as \sim 55 ms. This early frontal difference could be seen for all arrows except for the first arrow of the sequence (Figure S2A). Interestingly, there was a behavioral counterpart of this effect: whereas the first arrow had equal effects on the decision for LV and HV arrows, there were large differences in the weight of the subsequent arrows on the decision (Figure 1F). This suggests that only after the visibility of the sequence was established, on the basis of the first arrow, did participants treat the incoming information differently for LV and HV arrows. We speculate therefore that this frontal amplification may be a source of the behaviorally observed biasing effect [41].

While a change in evidence increased activity over parietal and central areas for both HV and LV evidence (albeit weaker, Figure 4), a neural influence of accumulated evidence on the processing of the current arrow was again found only for HV evidence. This MEG observation corroborates earlier behavioral and neural results [20] and suggests a neural implementation of the biasing of later information by past visible information, namely by a late (~200–300 ms after stimulus onset) top-down modulation of sensory representations (Figure 3).

By manipulating the configuration of the mask only, we created large differences in stimulus visibility without introducing differences in stimulus strength [29], as evidenced by equal priming effects under LV and HV conditions when a single arrow was presented (Figure 1K). Given that priming was unrelated to stimulus awareness, it is quite remarkable that the accumulation of evidence was. What may underlie these differences? Direct automatic priming effects are probably mediated by fast feedforward activations, which directly influence the evolving motor decision program [10]. These feedforward activations are relatively "automatic" [35] and have been found to be unaffected by stimulus visibility [14], although they can be modulated by several top-down factors, such as attention [42,43] and task-set [44]. In contrast, the slow accumulation of evidence over time, as probed in the present study, may require self-sustainable recurrent interactions between distant brain regions, which may only be

present when participants have complete access to (i.e., full awareness of) the stimuli [19].

Previous studies have shown that subliminal information can be accumulated linearly over a few hundreds of milliseconds [13,14,45-47]. Although indirect consequences of subliminal information can be measured for several minutes [17] and up to even as long as 24 h after its presentation [18], these effects may betray a form of learning and therefore synaptic changes rather than long-lasting subliminal activation. Indeed, most priming studies reveal a fast decay of subliminal information within less than one second [48,49]. Relative to this state of knowledge, the current study is the first to show that information from sequentially presented masked stimuli can be accumulated linearly over a long duration of more than a second. However, we also show a qualitative difference in how evidence is treated by the nervous system depending on the level of sensory awareness. As noted above, this qualitative difference need not imply that the processing pathway is entirely different for HV compared to LV trials. Rather, the same decision mechanism may be involved, with the main difference being that only a trickle of evidence is accumulated on LV trials, with the consequence that the decision threshold is typically not reached, therefore preventing the subsequent deployment of top-down strategies for down-weighting further incoming arrows. Indeed, our results suggest that the parietal and prefrontal regions that implement such decision making by evidence accumulation [1,2] may integrate sensory evidence across long periods of time, whether or not the original information was above or below the threshold for conscious access, but with a much weaker signal in the latter case.

A similar conclusion was reached by Sackur and Dehaene [50] when studying sequential two-step tasks with subliminal versus visible digits. As here, a qualitative behavioral difference was seen: participants were only able to perform a chained task of addition followed by comparison when the target digits were consciously seen, although they could perform each individual computation above chance when the digits were subliminal. This difference, although qualitative, could have arisen from the fact that subliminal digits did not yield enough evidence to ever reach threshold for the first computational step of the chained task. Thus, as in the present case, the same processing chain could have been in place on both conscious and non-conscious trials, but with non-conscious stimuli providing much smaller evidence that did not allow participants to go past the first processing stage and deploy further strategies.

There has been ample speculation about the function of awareness, ranging from none whatsoever [51,52] to enabling social communication [53]. Our results suggest a potential role of awareness in biasing information processing, namely the strategic exploitation of information in line with prior expectations and goals. This proposal fits with earlier hypotheses which link conscious access with flexible information processing, owing to the possibility of quickly circulating the conscious information to virtually all of the brain's higher level processors [54–57]. It also fits with a role of consciousness in enabling "meta-cognition," the ability to introspect about self-performance, which also has been associated with high-level processing in the prefrontal cortex [58]. Here, this strategic biasing process showed clear behavioral and neural advantages: it sped up processing and reduced neural computations related to the decision process when enough evidence had already been accrued. Under conditions of severely degraded evidence (such as near-threshold or subliminal information), the most rational strategy could, however, be to give each piece of evidence equal weight [20]. Interestingly, the strategic biasing process for highly visible information may exactly be the reason why "conscious" decision-making may in some special cases actually be poorer than "unconscious" decision-making [59,60], namely when an unbiased (equal) weighting of the evidence is required.

Materials and Methods

Participants

All participants in all experiments had normal or corrected-tonormal vision. The study was approved by the local institutional review board (CMO Arnhem-Nijmegen), and a written informed consent was obtained from the participants according to the Declaration of Helsinki, explicating that they agreed to participate in the MEG and behavioral experiments.

Stimuli

The experimental stimuli in all experiments were leftward and rightward pointing arrows. Stimuli were black, presented on a grey background, and subtended visual angles of $2.0^{\circ} \times 0.87^{\circ}$ (see Figure S1). Stimuli were presented using a PC running Presentation software (Neurobehavioral systems, Albany, USA) and shown on a screen that was ~75 cm away from the participant. Mask stimuli were constructed such as to either substantially reduce visibility of the stimuli (metacontrast mask), leading to low-visibility (LV) stimuli, or have only weak masking properties (pseudo mask), leading to high-visibility (HV) stimuli. Masks were identical in terms of overall luminance.

Experiment 1: Decision-Making Task

Sixteen healthy participants (5M/11F, age range 23-35) participated in the decision-making task (640 trials) within the MEG environment. Participants were presented with sequences of five successive arrows, each of which were briefly presented (17 ms), and followed 50 ms after its onset by a mask (100 ms), and a blank (150 ms). Therefore, the stimulus onset asynchrony (SOA) between successive arrows was 300 ms. Half of the trials contained metacontrast masks (leading to low-visibility [LV] stimuli) and the other half pseudo-masks (leading to high-visibility [HV] stimuli; see Figure 1A). Each arrow sequence contained either all LV or all HV arrows. At the end of each arrow sequence, the fixation square turned green, and the participants had to decide as quickly as possible whether the predominant direction of the arrow stimuli was left or right, by pressing a button with their left or right hand. Participants had to respond within a 500 ms time window. Each trial was followed by a baseline interval, during which a red fixation square was displayed for an average duration of 2,000 ms (between 1,750 and 2,250 ms). Several days before the MEG experiment participants were invited to the lab day to practice the task (~ 0.5 h). Prior to MEG data acquisition, participants engaged in an additional brief training session. During MEG data acquisition, participants engaged in 10 task blocks, each block consisting of 64 trials. Total duration of the experiment was ~ 60 min. For five participants, we collected only eight task blocks, due to time constraints.

For the analysis of reaction times (RT) and responses, we discarded trials to which participants responded very early (RT<150 ms), after the reaction time cut-off (RT>500 ms) or not at all (missed trials). For the analysis of responses, we compared the proportion of left/right responses as a function of the amount of accumulated evidence for a left/right response, for HV and LV trials (Figure 1D). For reaction times, we compared the RTs as a function of accumulated evidence for HV and LV trials (Figure 1E). For the analysis of arrow impact as a function of time, we used a logistic multiple regression analysis, in order to

independently estimate the effect of each arrow on the decision (Figure 1F). For the analysis of arrow influence as a function of previously accumulated evidence, we quantified the change in proportion of left/right response as a function of the direction of the last arrow, for the three levels of previously accumulated evidence (0, 2, and 4; see Figure 1B and Figure 1G). To investigate (differences in) linear trends as a function of accumulated evidence or time, we performed linear regression analysis for each participant and tested the significance of (differences in) slopes using (paired samples) t tests.

We recorded ongoing brain activity during Experiment 1 using a whole-head MEG with 275 axial gradiometers (VSM/CTF Systems, Port Coquitlam, British Columbia, Canada). Head localization was monitored continuously during the experiment using coils that were placed at the cardinal points of the head (nasion, left and right ear canal). The magnetic fields produced by these coils were used to measure the position of the participant's head with respect to the MEG sensor array. In addition to the MEG, the electrooculogram (EOG) was recorded from the supraorbital and infraorbital ridge of the left eye for the subsequent artifact rejection.

All data analysis was performed using the FieldTrip toolbox developed at Donders Institute for Brain, Cognition and Behaviour [61] using Matlab 7 (MathWorks, Natick, MA, USA). Data were checked for artifacts using a semiautomatic routine that helped detecting and rejecting eye blinks, muscle artifacts, and jumps in the MEG signal caused by the SQUID electronics. Subsequently, independent component analysis was used to remove any heart artifacts and eye movements not rejected by the semiautomatic routine. Finally, we low-pass filtered the data using a two-pass Butterworth filter (filter order of 6, frequency cutoff of 40 Hz). We did not apply any high-pass filter. We calculated an estimate of the planar gradient for the data analysis on the sensor level. The horizontal and vertical components of the planar gradients were calculated for each sensor using the signals from the neighboring sensors, thus approximating the signal measured by MEG systems with planar gradiometers. The planar field gradient simplifies the interpretation of the sensor-level data because the maximal signal typically is located above the source [62]. We established the significance of the differences in field strength for each experimental factor at the cluster level, using a nonparametric cluster randomization test. This test effectively controls the type I error rate in situations involving multiple comparisons (such as 275 sensors) by clustering neighboring sensor pairs that exhibit the same effect. The randomization method first identified sensors whose t statistics exceeded a critical value when comparing two conditions sensor by sensor (p < 0.05, two-sided). In the second step, to correct for multiple comparisons, contiguous sensors (separated by <5 cm) that exceeded the critical value (as defined in the first step) were considered a cluster. The cluster-level test statistic was defined from the sum of the t values of the sensors in a given cluster. The cluster with the maximum sum was used in the test statistics. The type I error rate for the complete set of 275 sensors was controlled by evaluating the cluster-level test statistic under the randomization null distribution of the maximum clusterlevel test statistic. This was obtained by randomizing the data between the two conditions across multiple participants, calculating t statistics for the new set of clusters. A reference distribution of cluster-level t statistics was created from 1,000 randomizations. The p value was estimated according to the proportion of the randomization null distribution exceeding the observed maximum cluster-level test statistic (the so-called Monte Carlo p value). MEG data analysis focused on (1) overall differences between processing of LV and HV information; (2) neural markers of accumulated

evidence for LV and HV information; and (3) effects of change in evidence (i.e., repeated versus different arrow direction) for LV and HV information. In all cases, we performed statistical tests (corrected for multiple comparisons) at five intervals after the onset of the arrow stimulus (from 50-300 ms in 50 ms steps). The first 50 ms after the onset of each arrow stimulus were used as a baseline interval. This "baseline" interval was physiologically motivated, for it takes approximately 50 ms for a visual stimulus to reach the cortex [63]. The aim of this baseline procedure was to effectively remove spill-over of overall activity from the previous arrow by subtracting out the activity at the onset of the arrow stimulus. A caveat of this procedure is that the previous LV/HV arrow may lead to a late (>350 ms) difference in evoked activity, which is misinterpreted as early differential activity evoked by a later arrow. Inspection of non-baseline-corrected traces suggests that this was not the case for our data (see Figure S2A, lower panel), but this possibility can nevertheless not be conclusively ruled out. For the analysis of overall differences between LV and HV arrows, we compared activity during LV and HV arrows, averaged across all five arrow presentations. For the analysis of global effects of accumulation evidence, we compared activity for LV and HV arrows that had "low prior accumulated evidence" and "high prior accumulated evidence," averaged across the third to fifth arrow presentation (since there is no differential amount of prior accumulated evidence until after the first two arrows are presented). Low evidence consisted of trials with zero (for third and fifth arrow) or one (for fourth arrow) prior accumulated evidence at the onset of the arrow. High evidence consisted of trials with two (for third and fifth arrow), three (for fourth arrow), or four (for fifth arrow) prior accumulated evidence at the onset of the arrow. For the analysis of the effect of change in evidence, we compared activity for LV and HV arrows that were either preceded by the same arrow (repeat) or preceded by the opposite arrow (change), averaged across the second to fifth arrow presentation (since there is no preceding arrow until after the first arrow is presented).

We also sought to establish the first time point of significant differences between LV and HV stimuli for the three clusters that showed a significant difference between stimulus types (frontocentral, parietal, and occipital clusters). We carried out *t* tests on 5 ms time intervals for the 50–300 ms interval after stimulus onset, on the difference wave between HV and LV stimuli, for each cluster. We defined the first time point of significant difference between HV and LV stimuli as the first sample in which five contiguous samples (i.e., 25 ms) showed a significant (p<0.05, two-tailed) difference between conditions.

Experiment 2: Visibility Task

All participants of Experiment 1 also participated in Experiment 2, while still in the MEG environment. To test visibility of strongly and weakly masked arrows, participants engaged in a six-choice discrimination task (120 trials, 50% LV and 50% HV). Stimulus and trial timing was exactly the same as in the experimental task with the exception that after the presentation of a trial, the question "How many arrows were pointing to the left/right?" was presented. This question remained on the screen until the participant made a response, after which a new trial started. Participants had to indicate their decision by pressing one of six response buttons. Whether participants were instructed to detect right- or left-pointing arrows was counterbalanced across participants. Before administering this task, participants were told that only accuracy was important in this task, not the speed of responding. To minimize strategic guessing, participants were notified of the fact that in this task, overall, equal numbers of trials (10) of each evidence level were presented.

Experiment 3: Confidence Task

Sixteen participants (5M/11F), age range 20–32), who did not participate in Experiments 1 and 2, took part in the confidencerating task (512 trials). Here, we assessed the relationship between decision-making performance and subjective confidence. Stimulus parameters and timing were all identical to Experiment 1, with the exception of an additional confidence question at the end of each trial, after the participant had given his/her response. The following question was presented, 1 s after the participants' response: "How confident are you about your response?" This sentence remained on the screen until the participant made a response, after which a new trial started. Participants had to indicate the confidence in their decision by pressing one of six buttons on the keyboard (1 = "pure guess", 6 = "100% sure"). The confidence response was unspeeded.

Experiment 4: Masked Priming Task

All participants of Experiment 3 also participated in Experiment 4. Here we assessed the amount of priming engendered by LV and HV arrows, using a masked priming experiment (640 trials). Here, the outline of the mask also formed an arrow stimulus (see Figure S1). Participants were instructed to respond as fast as possible to the direction of the mask arrow while ignoring the preceding prime arrow. Stimulus duration was the same as in Experiment 1. Each trial was followed by a baseline interval with an average duration of 1,000 ms (between 750 and 1,250 ms).

Supporting Information

Figure S1 Stimuli used in the experiments. Arrow stimulus. The arrow stimuli subtended a visual angle of 2.0° by 0.87° in all experiments. For Experiments 1–3, we used rectangular mask stimuli. Mask stimuli were constructed such as to either have only weak masking properties (pseudo mask), leading to high-visibility (HV) stimuli, or substantially reduce visibility of the stimuli (metacontrast mask), leading to low-visibility (LV) stimuli. Masks were identical in terms of overall luminance. For the masked

References

- Gold JI, Shadlen MN (2007) The neural basis of decision making. Annu Rev Neurosci 30: 535–574.
- Heekeren HR, Marrett S, Ungerleider LG (2008) The neural systems that mediate human perceptual decision making. Nat Rev Neurosci 9: 467–479.
- Shadlen MN, Newsome WT (2001) Neural basis of a perceptual decision in the parietal cortex (area LIP) of the rhesus monkey. J Neurophysiol 86: 1916–1936.
 Yang T, Shadlen MN (2007) Probabilistic reasoning by neurons. Nature 447:
- 1075-1080.
 Bar M, Biederman I (1998) Subliminal visual priming. Psychological Science 9: 464-468.
- Bar M, Biederman I (1999) Localizing the cortical region mediating visual awareness of object identity. Proc Natl Acad Sci U S A 96: 1790–1793.
- Bar M, Tootell RB, Schacter DL, Greve DN, Fischl B, et al. (2001) Cortical mechanisms specific to explicit visual object recognition. Neuron 29: 529–535.
 Naccache L, Dehaene S (2001) Unconscious semantic priming extends to novel
- Naccache L, Dehaene S (2001) Unconscious semantic priming extends to novel unseen stimuli. Cognition 80: 215–229.
 Dehaene S, Naccache L, Cohen L, Bihan DL, Mangin IF, et al. (2001) Cerebral
- Dehaene S, Naccache L, Cohen L, Bihan DL, Mangin JF, et al. (2001) Cerebral mechanisms of word masking and unconscious repetition priming. Nat Neurosci 4: 752–758.
- Dehaene S, Naccache L, Le Clec HG, Koechlin E, Mueller M, et al. (1998) Imaging unconscious semantic priming. Nature 395: 597–600.
- Lau HC, Passingham RE (2007) Unconscious activation of the cognitive control system in the human prefrontal cortex. J Neurosci 27: 5805–5811.
- van Gaal S, Ridderinkhof KR, Scholte HS, Lamme VA (2010) Unconscious activation of the prefrontal no-go network. J Neurosci 30: 4143–4150.
- Del Cul A, Baillet S, Dehaene S (2007) Brain dynamics underlying the nonlinear threshold for access to consciousness. PLoS Biol 5: e260. doi:10.1371/ journal.pbio.0050260.

priming task (Experiment 4), the outline of the mask also formed an arrow stimulus. (EPS)

Figure S2 Time course of activity (differences) for each arrow. (A) Time course of overall differences in neural processing of lowand high-visibility arrow stimuli, for the fronto-central, parietal, and occipital cluster outlined in Figure 2. Shown are baselinecorrected traces (top panel) as well as non-baseline-corrected traces (bottom panel). A direct comparison of these shows that there is a superposition of activity for each subsequent arrow stimulus as well as an overall activity increase, which is strongest over frontal sensors, and somewhat visible over occipital and parietal sensors. By removing the baseline differences, the signal reflects the increase in activity generated by the current stimulus, over and above the activity generated by earlier arrows. (B) Time course of global effect of accumulated evidence for "low evidence" and "high evidence" arrows, for low- and high-visibility arrow stimuli, in the clusters outlined in Figure 3. (C) Time course of activity for "repeat" and "change" arrows, for low- and high-visibility arrow stimuli, in the clusters outlined in Figure 4.

(EPS)

Figure S3 Neural markers of change of evidence for nine "poor perceivers" who scored at chance level in the six-choice discrimination task. Topographical representation of larger activity (between 150 and 250 ms) for arrow stimuli that were different from their directly preceding arrows ("change"), compared to arrow stimuli which were the same direction as their directly preceding arrows ("repeat"). This activity difference was plotted for high-visibility (top row) and low-visibility (middle row) arrows, as well as the difference in activity between them (bottom row). Differences are robust and qualitatively similar to those in the whole group (see Figure 4 for whole group results). (EPS)

Author Contributions

The author(s) have made the following declarations about their contributions: Conceived and designed the experiments: FdL SvG SD. Performed the experiments: FdL SvG. Analyzed the data: FdL SvG. Wrote the paper: FdL SvG VL SD.

- Vorberg D, Mattler U, Heinecke A, Schmidt T, Schwarzbach J (2003) Different time courses for visual perception and action priming. Proc Natl Acad Sci U S A 100: 6275–6280.
- van Gaal S, Lamme VA, Ridderinkhof KR (2010) Unconsciously triggered conflict adaptation. PLoS One 5: e11508. doi:10.1371/journal.pone.0011508.
- Bodner GE, Mulji R (2010) Prime proportion affects masked priming of fixed and free-choice responses. Exp Psychol 57: 360–366.
- Gaillard R, Cohen L, Adam C, Clemenceau S, Hasboun D, et al. (2007) Subliminal words durably affect neuronal activity. Neuroreport 18: 1527–1531.
- Sweeny TD, Grabowecky M, Suzuki S, Paller KA (2009) Long-lasting effects of subliminal affective priming from facial expressions. Conscious Cogn 18: 929–938.
- Dehaene S, Changeux JP, Naccache L, Sackur J, Sergent C (2006) Conscious, preconscious, and subliminal processing: a testable taxonomy. Trends Cogn Sci 10: 204–211.
- de Lange FP, Jensen O, Dehaene S (2010) Accumulation of evidence during sequential decision making: the importance of top-down factors. J Neurosci 30: 731–738.
- Kiani R, Hanks TD, Shadlen MN (2008) Bounded integration in parietal cortex underlies decisions even when viewing duration is dictated by the environment. J Neurosci 28: 3017–3029.
- Beck JM, Ma WJ, Kiani R, Hanks T, Churchland AK, et al. (2008) Probabilistic population codes for Bayesian decision making. Neuron 60: 1142–1152.
- van Aalderen-Smeets S, Oostenveld R, Schwarzbach J (2005) Investigating neurophysiological correlates of metacontrast masking with magnetoencephalography. Advances in Cognitive Psychology 2: 21–35.
- Lau HC (2008) A higher order Bayesian decision theory of consciousness. Progress in Brain Research 168: 35–48.

- Nisbett R, Wilson T (1977) Telling more than we can know. Psychological Review 84: 231–295.
- Oliveira FT, Diedrichsen J, Verstynen T, Duque J, Ivry RB (2010) Transcranial magnetic stimulation of posterior parietal cortex affects decisions of hand choice. Proc Natl Acad Sci U S A 107: 17751–17756.
- Stoloff RH, Taylor JA, Xu J, Ridderikhoff A, Ivry RB (2011) Effect of reinforcement history on hand choice in an unconstrained reaching task. Front Neurosci 5: 41.
- Persaud N, Lau H (2008) Direct assessment of qualia in a blindsight participant. Conscious Cogn 17: 1046–1049.
- Francken J, Van Gaal S, de Lange FP (2011) Immediate and long-term priming effects are independent of prime awareness. Conscious Cogn In press.
- Kayser AS, Buchsbaum BR, Erickson DT, D'Esposito M (2010) The functional anatomy of a perceptual decision in the human brain. J Neurophysiol 103: 1179–1194.
- Melloni L, Schwiedrzik CM, Muller N, Rodriguez E, Singer W (2011) Expectations change the signatures and timing of electrophysiological correlates of perceptual awareness. J Neurosci 31: 1386–1396.
- Gao J, Wong-Lin K, Holmes P, Simen P, Cohen JD (2009) Sequential effects in two-choice reaction time tasks: decomposition and synthesis of mechanisms. Neural Comput 21: 2407–2436.
- Dehaene S, Jobert A, Naccache L, Ciuciu P, Poline JB, et al. (2004) Letter binding and invariant recognition of masked words: behavioral and neuroimaging evidence. Psychol Sci 15: 307–313.
- Ratcliff R, McKoon G (2008) The diffusion decision model: theory and data for two-choice decision tasks. Neural Comput 20: 873–922.
- Gaillard R, Dehaene S, Adam C, Clemenceau S, Hasboun D, et al. (2009) Converging intracranial markers of conscious access. PLoS Biol 7: e61. doi:10.1371/journal.pbio.1000061.
- Lamme VA, Zipser K, Spekreijse H (2002) Masking interrupts figure-ground signals in V1. J Cogn Neurosci 14: 1044–1053.
- Macknik SL, Livingstone MS (1998) Neuronal correlates of visibility and invisibility in the primate visual system. Nat Neurosci 1: 144–149.
- Lamme VA, Roelfsema PR (2000) The distinct modes of vision offered by feedforward and recurrent processing 1. Trends Neurosci 23: 571–579.
- Enns JT, Lleras A (2008) What's next? New evidence for prediction in human vision. Trends Cogn Sci 12: 327–333.
- Di Lollo V, Enns JT, Rensink RA (2000) Competition for consciousness among visual events: the psychophysics of reentrant visual processes. J Exp Psychol Gen 129: 481–507.
- Bar M, Kassam KS, Ghuman AS, Boshyan J, Schmid AM, et al. (2006) Topdown facilitation of visual recognition. Proc Natl Acad Sci U S A 103: 449–454.
- Sumner P, Tsai PC, Yu K, Nachev P (2006) Attentional modulation of sensorimotor processes in the absence of perceptual awareness. Proc Natl Acad Sci U S A 103: 10520–10525.

- Naccache L, Blandin E, Dehaene S (2002) Unconscious masked priming depends on temporal attention. Psychol Sci 13: 416–424.
- Nakamura K, Dehaene S, Jobert A, Le Bihan D, Kouider S (2007) Task-specific change of unconscious neural priming in the cerebral language network. Proc Natl Acad Sci U S A 104: 19643–19648.
- Jaskowski P, Skalska B, Verleger R (2003) How the self controls its "automatic pilot" when processing subliminal information. J Cogn Neurosci 15: 911–920.
- Frings C, Bermeitinger C, Wentura D (2008) Center-surround or spreading inhibition: which mechanism caused the negative effect from repeated masked semantic primes? Exp Psychol 55: 234–242.
- Wentura D, Frings C (2005) Repeated masked category primes interfere with related exemplars: new evidence for negative semantic priming. J Exp Psychol Learn Mem Cogn 31: 108–120.
- Greenwald AG, Draine SC, Abrams RL (1996) Three cognitive markers of unconscious semantic activation. Science 273: 1699–1702.
- Dupoux E, de Gardelle V, Kouider S (2008) Subliminal speech perception and auditory streaming. Cognition 109: 267–273.
- Sackur J, Dehaene S (2009) The cognitive architecture for chaining of two mental operations. Cognition 111: 187–211.
- 51. Flanagan O (1992) Consciousness reconsidered. Cambridge, MA: MIT Press.
- Lau H, Rosenthal D (2011) Empirical support for higher-order theories of conscious awareness. Trends Cogn Sci 15: 365–373.
- 53. Frith CD, Frith U (2007) Social cognition in humans. Curr Biol 17: R724–R732.
- Shallice T (1988) From neuropsychology to mental structure. New York: Cambridge University Press.
- Posner MI, Rothbart MK (1998) Attention, self-regulation and consciousness. Philos Trans R Soc Lond B Biol Sci 353: 1915–1927.
- Baars BJ (1989) A cognitive theory of consciousness. Cambridge, UK: Cambridge University Press.
- Dehaene S, Naccache L (2001) Towards a cognitive neuroscience of consciousness: basic evidence and a workspace framework. Cognition 79: 1–37.
 Floring SM Weil PS, Nacca Z, Dale PL, Pare C (2010) PL dependence and provide the second seco
- Fleming SM, Weil RS, Nagy Z, Dolan RJ, Rees G (2010) Relating introspective accuracy to individual differences in brain structure. Science 329: 1541–1543.
 Voss IL, Bayn CL, Paller KA (2008) Accurate forced-choice recognition without
- Voss JL, Baym CL, Paller KA (2008) Accurate forced-choice recognition without awareness of memory retrieval. Learn Mem 15: 454–459.
 Dijksterhuis A, Bos MW, Nordgren LF, van Baaren RB (2006) On making the
- right choice: the deliberation-without-attention effect. Science 311: 1005–1007.
- Oostenveld R, Fries P, Maris E, Schoffelen JM (2011) FieldTrip: Open source software for advanced analysis of MEG, EEG, and invasive electrophysiological data. Comput Intell Neurosci 2011: 156869.
- Bastiaansen MC, Knosche TR (2000) Tangential derivative mapping of axial MEG applied to event-related desynchronization research. Clin Neurophysiol 111: 1300–1305.
- Nowak LG, Munk MH, Girard P, Bullier J (1995) Visual latencies in areas V1 and V2 of the macaque monkey. Vis Neurosci 12: 371–384.